Author’s response to reviews

Title: Comparison of different rating scales for the use in Delphi studies: Different scales lead to different consensus and show different test-retest reliability

Authors:

Toni Lange (toni.lange@uniklinikum-dresden.de)
Christian Kopkow (Christian.Kopkow@hs-gesundheit.de)
Jörg Lützner (Joerg.Luetzner@uniklinikum-dresden.de)
Klaus-Peter Günther (Klaus-Peter.Guenther@uniklinikum-dresden.de)
Sascha Gravius (Sascha.Gravius@ukb.uni-bonn.de)
Hanns-Peter Scharf (Hanns-Peter.Scharf@umm.de)
Johannes Stöve (Johannes.Stoeve@st-marienkrankenhaus.de)
Richard Wagner (Richard.Wagner@fdk.info)
Jochen Schmitt (Jochen.Schmitt@uniklinikum-dresden.de)

Version: 5 Date: 20 Jan 2020

Author’s response to reviews:

Dear Sarah Gray and Rebecca Strong

Thank you very much for accepting our paper for publication in the BMC Medical Research Methodology. We are delighted that the paper was positively reviewed.

Kindest regards

Toni Lange on behalf of all co-authors